Demand-Aware Beam Hopping and Power Allocation for Load Balancing in Digital Twin empowered LEO Satellite Networks

Ruili Zhao, Student Member, IEEE, Jun Cai, Senior Member, IEEE, Jiangtao Luo, Senior Member, IEEE, Junpeng Gao, Student Member, IEEE and Yongyi Ran, Member, IEEE,

Abstract—Low-Earth orbit (LEO) satellites utilizing beam hopping (BH) technology offer extensive coverage, low latency, high bandwidth, and significant flexibility. However, the uneven geographical distribution and temporal variability of ground traffic demands, combined with the high mobility of LEO satellites, present significant challenges for efficient beam resource utilization. Traditional BH methods based on GEO satellites fail to address issues such as satellite interference, overlapping coverage, and mobility. This paper explores a Digital Twin (DT)based collaborative resource allocation network for multiple LEO satellites with overlapping coverage areas. A two-tier optimization problem, focusing on load balancing and cell service fairness, is proposed to maximize throughput and minimize intercell service delay. The DT layer optimizes the allocation of overlapping coverage cells by designing BH patterns for each satellite, while the LEO layer optimizes power allocation for each selected service cell. At the DT layer, an Actor-Critic network is deployed on each agent, with a global critic network in the cloud center. The A3C algorithm is employed to optimize the DT layer. Concurrently, the LEO layer optimization is performed using a Multi-Agent Reinforcement Learning algorithm, where each beam functions as an independent agent. The simulation results show that this method reduces satellite load disparity by about 72.5% and decreases the average delay to 12ms. Additionally, our approach outperforms other benchmarks in terms of throughput, ensuring a better alignment between offered and requested data.

Index Terms—Multi-satellite beam hopping, digital twin, power allocation, load balancing, LEO satellite communications.

I. INTRODUCTION

OW Earth Orbit (LEO) satellite communication systems, which play a crucial role in 6G, have advanced quickly due to their wide coverage, low latency, high transmission capacity, and flexible resource scheduling [1]–[4]. To provide broadband transmission and seamless coverage, LEO

This work is jointly supported by National Science Foundation of China (No.62171072,62172064,62003067), Chongqing Postgraduate Research and Innovation Project (CYB21204) and the China Scholarship Council program (Project ID: 202308500242). (Corresponding authors: Jun Cai).

Ruili Zhao and Junpeng Gao are with the School of Communication and Information Engineering, Chongqing University of Posts and Telecommunications, Chongqing 400065, China (e-mail: d200101029@stu.cqupt.edu.cn; d200101007@stu.cqupt.edu.cn).

Jiangtao Luo and Yongyi Ran are with the Electronic Information and Networking Research Institute, Chongqing University of Posts and Telecommunications, Chongqing 400065, China (e-mail: luojt@cqupt.edu.cn; ranyy@cqupt.edu.cn).

Jun Cai is with the Network Intelligence and Innovation Laboratory (NI2L), Department of Electrical and Computer Engineering, Concordia University, Montreal, QC H3G 1M8, Canada (e-mail: jun.cai@concordia.ca). constellations must ensure multiple coverage, where some areas are served by multiple satellites. However, the uneven geographic distribution of ground users and the high mobility of LEO satellites lead to temporal and spatial variability in traffic demands within a satellite's coverage area. Beam Hopping (BH) technology, based on time slicing, achieves the traditional multi-beam coverage with fewer beams. This makes LEO satellites equipped with BH antennas highly flexible, which is crucial for addressing the varying demands across different regions [5]. Nonetheless, if beams are powered uniformly, the transmission capacity will still fall short of meeting the uneven traffic demands across different beams [6]-[8]. Additionally, onboard power in LEO satellites is extremely limited [9], making the intelligent and real-time joint BH and power allocation (PA) a critical research focus in LEO satellite communication systems.

Nevertheless, there are still some critical issues which are closely related to practical implementations and have not been well addressed in effectively scheduling beams and power resources in LEO satellite constellations [6], [8]. First, the constantly changing topology of LEO satellite networks leads to fluctuations in inter-beam interference and satellite-toground link conditions, demanding adaptive rather than static resource management [10], [11]. Second, managing resource scheduling is key to reducing beam interference, both within and between satellites. Third, joint optimization of beam and power resources expands the system's state-action space, exacerbating the "curse of dimensionality." Additionally, this joint optimization creates a discrete-continuous hybrid action space, where beam allocation is discrete, and PA is continuous [12]. Approximating continuous actions using finite discrete sets diminishes the natural structure of continuous actions, while relaxing discrete actions into continuous sets significantly complicates the action space. Finally, scheduling beams and power resources must not be viewed as a simple single-objective optimization, but should concurrently consider multiple performance metrics, such as network throughput, latency, and fairness.

In reviewing the literature, it is evident that most studies concentrated on flexible radio resource allocation with a fixed beam direction, applicable to either a single satellite [6], [13]– [15] or multiple satellites [16]–[19]. Some studies (e.g., [20], [21]) primarily utilize BH technology to address the uneven geographical distribution of users. However, these studies overlook the importance of adaptive beam resource allocation, opting instead for a static allocation strategy, such as equal distribution across beams. Although certain works [22]–[24] are dedicated to joint optimization of PA and beam direction, they are limited to single-satellite scenarios.

In summary, existing research cannot be directly applied to multi-beam, multi-LEO satellite networks to address the aforementioned key issues. Furthermore, addressing all of these issues simultaneously presents new challenges for the dynamic resource management in LEO satellite networks due to the following aspects.

- a. Inter-beam interference is unavoidable when multi-beam LEO satellites transmit downlink data through BH. This interference includes both intra-satellite and inter-satellite interference, leading to a strong interdependence between resource allocation decisions across satellites. Therefore, to fully exploit the spectrum reuse benefits in multibeam LEO satellite communications, fostering cooperation among satellites is crucial to mitigate interference and enhance system capacity. However, the limited computational capacity and energy resources of satellites significantly increase the difficulty of satellite cooperation.
- **b.** The resource allocation decisions (including BH and PA) should be jointly optimized to maximize overall network performance. Since BH and PA are closely intertwined, their joint optimization is typically NP-hard, making it difficult to achieve an optimal solution within polynomial time.
- c. With the constant movement of LEO satellites, the connections between satellites and ground terminals frequently change. Consequently, resource allocation decisions must be adaptable to this dynamic environment, further increasing the complexity of resource management.

To address these complex challenges, we note that Digital Twin (DT) technology provides an opportunity to represent the real world in a virtualized manner and has emerged as a promising tool for guiding resource deployment and scheduling in the real world. As an effective method for bridging the gap between physical entities and the digital realm, DT has garnered significant attention from both industry and academia [25]. In industries such as manufacturing, power grids, and transportation management, DT has proven its ability to enhance operational efficiency and reduce resource waste through precise system modeling and predictive capabilities [26]. In multi-LEO satellite communication systems, introducing DT can provide a global perspective for LEO satellite systems. By virtually replicating and monitoring the status and resource dynamics of physical satellite networks, DT can fully capture the real-time status changes of satellites, ground users, and the environment [27]. This global perspective can support the overall optimization of BH and PA, not only helping to coordinate resource allocation for multiple satellites, but also effectively dealing with interference management and load balancing issues between satellites.

Motivated by these considerations, we design a DTempowered multi-beam, multi-satellite cooperative service network and formulate a resource management problem for the downlink scenario that integrates both BH and PA. Our objective is to enhance the fairness of cell service while simultaneously improving throughput. To address this esource management problem, we decompose the original problem into two sub-problems: demand prediction-based BH at the DT layer and beam PA at the LEO satellite layer. We propose two advanced reinforcement learning algorithms to solve these subproblems. The key contributions of this paper are summarized as follows:

- We propose a DT-empowered resource allocation framework for LEO satellite networks, addressing the joint BH and PA problem in a dynamic multi-beam, multisatellite downlink scenario. The objective is to maximize throughput while ensuring fairness across cells. Given the mixed-integer, non-convex nature of the optimization problem, we decompose it into two simpler sub-problems: a DT layer multi-LEO satellite cooperative BH problem focused on load balancing, and a multi-beam power resource allocation problem at the LEO layer.
- For optimization at the DT layer, the cloud predicts future time slot requirements based on historical demand data, considering the overlapping coverage of LEO satellite systems to design a BH scheme that balances communication demands among satellites. We employ the A3C algorithm, deploying an actor-critic network on each satellite and a global critic network in the cloud to improve training efficiency and stability.
- To optimize resource allocation at the LEO layer, each satellite beam is modeled as an agent, and an on-demand beam PA problem is established. The Multi-Agent Deep Deterministic Policy Gradient (MADDPG) algorithm is utilized to facilitate cooperation and competition among these agents, further enhancing cell service fairness and system throughput.
- Extensive simulation results demonstrate that the proposed joint BH and PA algorithm significantly outperforms baseline algorithms in terms of throughput, load balancing, and cell fairness.

The rest of this paper is organized as follows. Section II reviews the related literature. Section III describes the system model and presents the joint BH and power resource allocation optimization problem for multiple LEO satellites. Section IV decomposes the optimization problem into two sub-problems and provides a detailed explanation of the solution methods for these sub-problems. Section V presents the simulation results, demonstrating the superior performance of the proposed method. Section VI presents a summary of this paper and discusses future research prospects.

II. RELATED WORK

Recently, to better align beam resources with non-uniform traffic demands, flexible resource allocation strategies for satellite communication systems have garnered significant attention [6], [17], [18], [24], [28]–[46].

For instance, [6] presented a PA and multi-beam scheduling method to optimize limited power resources against unevenly distributed user traffic, considering the trade-off between total capacity and user fairness. [17] focused on subchannel and PA optimization in multi-beam GEO satellite communication systems, aiming to satisfy traffic demands with minimal transmission power and bandwidth. In [18], the authors formulated a joint optimization problem of PA and data transmission scheduling for satellite-assisted remote IoT networks, with the objective of maximizing total data rate, and they designed a model-free reinforcement learning framework to accommodate the highly dynamic nature of LEO satellite communications.

Similarly, Takahashi et al. [29] introduced a novel mathematical model that balances beam pointing and PA, significantly improving satellite communication resource utilization and highlighting the benefits of joint optimization over singledimensional optimization. Wang et al. [30] proposed a joint optimization algorithm for PA, beam scheduling, and terminal time-slot allocation in coexisting BH-NOMA systems, devising a constraint scheme to accurately gauge the global optimal solution. Du et al. [31] proposed a dynamic reinforcement learning-based strategy to integrate beam hopping and power allocation (BHPA), enhancing power utilization and satellite throughput while demonstrating the feasibility of PA.

Further, Shi et al. [34] addressed the multi-objective optimization problem of system throughput and terminal delay, employing a heuristic method to jointly allocate power and bandwidth. The authors in [38] developed a two-stage approach for beam PA, utilizing genetic algorithm and simulated annealing (GA-SA). Lin et al. [39] presented a two-stage BH design strategy aimed at minimizing energy consumption in BH satellite communications while addressing users' heterogeneous demands.

However, the aforementioned studies primarily focus on single GEO satellites. Joint BH and resource allocation for multiple LEO satellites presents distinct challenges. The rapid motion of LEO satellites, coupled with constantly changing channel conditions and varying traffic demands, differentiates the LEO scenario from GEO systems. Additionally, LEO terminals are typically small and prone to interference from neighboring satellites, especially in regions covered by multiple satellites. This makes it challenging to directly apply GEObased BH resource allocation schemes to LEO systems.

In response, recent research has begun investigating resource allocation strategies for multi-beam LEO satellite systems [40]–[46]. Liu et al. [40] modeled the dynamic coverage area of LEO satellites as rectangular blocks and applied an iterative algorithm to optimize system capacity. A greedy algorithm based on beam position traffic demand was utilized to enhance throughput [41]. Shuang et al. [42] introduced a joint beam scheduling and power optimization algorithm considering the geographic distribution of aggregation nodes, while examining the impact of co-channel interference in LEO systems, focusing primarily on single LEO satellites. Li [43] presented an optimization solution for the high-dimensional problem in dual-satellite scenarios. Deng [44] investigated system capacity optimization in three-layer heterogeneous satellite networks, focusing on load balancing and interference management. Lin et al. [45], [46] explored the BH scheme for NGSO multi-satellite systems and the coexistence of NGSO and GSO satellites. These studies, however, assume even power distribution across satellite beams, leaving opportunities for further optimization.

The most closely related work is [47], which explored the joint optimization of beam direction alongside spectrum, time, and power resources in dmicyna multi-beam LEO satellite networks. Nonetheless, the overlapping beam coverage of multiple LEO satellites introduces the risk of interference between adjacent satellites. Although beam direction control has been addressed, the issue of interference management in multi-satellite cooperation has not been fully resolved. This paper aims to tackle this problem comprehensively.

III. SYSTEM MODEL AND PROBLEM FORMULATION

This section first describes the system model, which consists of the digital twin network, traffic model, and communication model. Next, the joint optimization problem of BH and PA is formulated. For convenience, the detailed notations and definitions used in this paper are summarized in Table I.

TABLE I: Notations and Definitions

$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
CNumber of cells covered per satelliteKNumber of beams per satelliteBTotal satellite bandwidthTBH periodJDelay fairnessQLoad balancing α The weight between Load balancing and delay fairness β The weight between throughput and delay fairness β The weight between throughput and delay fairness β The set of cells under the coverage of satellite n T_{ttl} Time to live of a data packet in queue $p_{n,c}^{t}$ Arrival traffic in time slot t for cell c of satellite n $p_{n,c}^{t}$ Predicted of arrival traffic at slot t for cell c of satellite n λ_n^t The arrival rate of cell c covered by satellite n in slot t $d_{n,c}^{t}$ Data packets stored in queue c at slot t in satellite n $\phi_{n,t}^{n,t}$ Data packets stored in queue c for l time slots $\sigma_{n,t}^{n,t}$ The illumination status of cell c by satellite n in time slot t
KNumber of beams per satelliteBTotal satellite bandwidthTBH periodJDelay fairnessQLoad balancing α The weight between Load balancing and delay fairness β The weight between throughput and delay fairness β The weight between throughput and delay fairness γ The set of cells under the coverage of satellite n T_{ttl} Time to live of a data packet in queue $\rho_{n,c}^n$ Arrival traffic in time slot t for cell c of satellite n $\rho_{n,c}^n$ Predicted of arrival traffic at slot t for cell c of satellite n λ_n^t The arrival rate of cell c covered by satellite n in slot t $d_{n,c}^t$ Data packets stored in queue c for l time slots $\sigma_{n,t}^{n,t}$ Data packets stored in queue c by satellite n in time slot t $t_{n,c}^t$ The illumination status of cell c by satellite n in time slot t
BTotal satellite bandwidthTBH periodJDelay fairnessQLoad balancing α The weight between Load balancing and delay fairness β The weight between throughput and delay fairness β The weight between throughput and delay fairness β The weight between throughput and delay fairness β The set of cells under the coverage of satellite n T_{ttl} Time to live of a data packet in queue $\rho_{n,c}^{t}$ Arrival traffic in time slot t for cell c of satellite n $\rho_{n,c}^{t}$ Predicted of arrival traffic at slot t for cell c of satellite n λ_n^{t} The arrival rate of cell c covered by satellite n in slot t $d_{n,c}^{t}$ Data packets stored in queue c at slot t in satellite n $\phi_{n,t}^{c,l}$ Data packets stored in queue c for l time slots $x_{n,c}^{t}$ The illumination status of cell c by satellite n in time slot t
TBH periodJDelay fairnessQLoad balancing α The weight between Load balancing and delay fairness β The weight between throughput and delay fairness β The weight between throughput and delay fairness V_n The set of cells under the coverage of satellite n T_{ttl} Time to live of a data packet in queue $\rho_{n,c}^{t}$ Arrival traffic in time slot t for cell c of satellite n $\rho_{n,c}^{t}$ Predicted of arrival traffic at slot t for cell c of satellite n λ_n^{t} The arrival rate of cell c covered by satellite n in slot t $\lambda_n^{t,c}$ The total traffic stored in queue c at slot t in satellite n $\phi_{n,t}^{t,l}$ Data packets stored in queue c for l time slots $\pi_{n,c}^{t,l}$ The illumination status of cell c by satellite n in time slot t L_n^{t} The traffic load of satellite n at slot t
$ \begin{array}{lll} J & \mbox{Delay fairness} \\ Q & \mbox{Load balancing} \\ \alpha & \mbox{The weight between Load balancing and delay fairness} \\ \beta & \mbox{The weight between throughput and delay fairness} \\ \gamma_n & \mbox{The set of cells under the coverage of satellite } n \\ T_{ttl} & \mbox{Time to live of a data packet in queue} \\ \rho^{t}_{n,c} & \mbox{Arrival traffic in time slot } t \mbox{ for cell } c \mbox{ of satellite } n \\ \rho^{t}_{n,c} & \mbox{Arrival traffic at slot } t \mbox{ for cell } c \mbox{ of satellite } n \\ \rho^{t}_{n,c} & \mbox{Predicted of arrival traffic at slot } t \mbox{ for cell } c \mbox{ of satellite } n \\ \lambda^{t}_{n,c} & \mbox{The total traffic stored in queue } c \mbox{ at slot } t \mbox{ in satellite } n \\ \phi^{n,t}_{c,l} & \mbox{ Data packets stored in queue } c \mbox{ for } l \mbox{ time slots } t \\ \pi^{t}_{n,c} & \mbox{The illumination status of cell } c \mbox{ by satellite } n \mbox{ in time slot } t \\ L^{t}_{n} & \mbox{ The traffic load of satellite } n \mbox{ at slot } t \end{array} $
$\begin{array}{cccc} Q & \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \$
$ \begin{array}{lll} \alpha & & \text{The weight between Load balancing and delay fairness} \\ \beta & & \text{The weight between throughput and delay fairness} \\ V_n & & \text{The set of cells under the coverage of satellite } n \\ T_{ttl} & & \text{Time to live of a data packet in queue} \\ \rho_{n,c}^t & & \text{Arrival traffic in time slot } t \text{ for cell } c \text{ of satellite } n \\ \rho_{n,c}^t & & \text{Predicted of arrival traffic at slot } t \text{ for cell } c \text{ of satellite } n \\ \lambda_n^t & & \text{The arrival rate of cell } c \text{ covered by satellite } n \text{ in slot } t \\ d_{n,c}^{n,t} & & \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \phi_{c,l}^{n,t} & & \text{Data packets stored in queue } c \text{ for } l \text{ time slots } t \\ x_{n,c}^{n,c} & & \text{The illumination status of cell } c \text{ by satellite } n \text{ in time slot } t \\ L_n^t & & \text{The traffic load of satellite } n \text{ at slot } t \end{array} $
$ \begin{array}{lll} \alpha & & \text{The weight between Load balancing and delay fairness} \\ \beta & & \text{The weight between throughput and delay fairness} \\ V_n & & \text{The set of cells under the coverage of satellite } n \\ T_{ttl} & & \text{Time to live of a data packet in queue} \\ \rho_{n,c}^t & & \text{Arrival traffic in time slot } t \text{ for cell } c \text{ of satellite } n \\ \rho_{n,c}^t & & \text{Predicted of arrival traffic at slot } t \text{ for cell } c \text{ of satellite } n \\ \lambda_n^t & & \text{The arrival rate of cell } c \text{ covered by satellite } n \text{ in slot } t \\ d_{n,c}^{n,t} & & \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \phi_{c,l}^{n,t} & & \text{Data packets stored in queue } c \text{ for } l \text{ time slots } t \\ x_{n,c}^{n,c} & & \text{The illumination status of cell } c \text{ by satellite } n \text{ in time slot } t \\ L_n^t & & \text{The traffic load of satellite } n \text{ at slot } t \end{array} $
$ \begin{array}{ccc} V_n & \text{The set of cells under the coverage of satellite } n \\ T_{ttl} & \text{Time to live of a data packet in queue} \\ \rho_{n,c}^t & \text{Arrival traffic in time slot } t \text{ for cell } c \text{ of satellite } n \\ \rho_{n,c}^t & \text{Predicted of arrival traffic at slot } t \text{ for cell } c \text{ of satellite } n \\ \lambda_n^t & \text{The arrival rate of cell } c \text{ covered by satellite } n \text{ in slot } t \\ d_{n,c}^t & \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \phi_{n,t}^{n,t} & \text{Data packets stored in queue } c \text{ for } l \text{ time slots } t \\ x_{n,c}^t & \text{The illumination status of cell } c \text{ by satellite } n \text{ in time slot } t \\ L_n^t & \text{The traffic load of satellite } n \text{ at slot } t \\ \end{array} $
$ \begin{array}{ccc} T_{ttl} & \text{Time to live of a data packet in queue} \\ \rho_{n,c}^{t} & \text{Arrival traffic in time slot } t \text{ for cell } c \text{ of satellite } n \\ \rho_{n,c}^{t} & \text{Predicted of arrival traffic at slot } t \text{ for cell } c \text{ of satellite } n \\ \lambda_{n}^{t} & \text{The arrival rate of cell } c \text{ covered by satellite } n \text{ in slot } t \\ d_{n,c}^{t} & \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \phi_{n,t}^{n,t} & \text{Data packets stored in queue } c \text{ for } l \text{ time slots} \\ x_{n,c}^{t} & \text{The illumination status of cell } c \text{ by satellite } n \text{ in time slot } t \\ L_{n}^{t} & \text{The traffic load of satellite } n \text{ at slot } t \\ \end{array} $
$ \begin{array}{ccc} \rho_{n,c}^{t} & \text{Arrival traffic in time slot } t \text{ for cell } c \text{ of satellite } n \\ \hat{\rho}_{n,c}^{t} & \text{Predicted of arrival traffic at slot } t \text{ for cell } c \text{ of satellite } n \\ \lambda_{n}^{t} & \text{The arrival rate of cell } c \text{ covered by satellite } n \text{ in slot } t \\ d_{n,c}^{t} & \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \phi_{c,l}^{n,t} & \text{Data packets stored in queue } c \text{ for } l \text{ time slots} \\ The illumination status of cell c by satellite n in time slot t \\ L_{n}^{t} & \text{The traffic load of satellite } n \end{array} $
$ \begin{array}{ll} \hat{\rho}_{n,c}^t \\ \lambda_n^t \\ d_{n,c}^t \\ r_{n,c}^t \\ r_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ t_n^t \\ t_$
$ \begin{array}{ll} \hat{\rho}_{n,c}^t \\ \lambda_n^t \\ d_{n,c}^t \\ r_{n,c}^t \\ r_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ x_{n,c}^t \\ t_n^t \\ t_$
$ \begin{array}{c} d_{n,c}^{t} \\ \phi_{n,t}^{n,t} \\ \phi_{c,l}^{n,t} \\ t_{n,c}^{t} \end{array} \begin{array}{c} \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \text{Data packets stored in queue } c \text{ for } l \text{ time slots} \\ \text{The illumination status of cell } c \text{ by satellite } n \text{ in time slot } t \\ L_{n}^{t} \end{array} $
$ \begin{array}{c} d_{n,c}^{t} \\ \phi_{n,t}^{n,t} \\ \phi_{c,l}^{n,t} \\ t_{n,c}^{t} \end{array} \begin{array}{c} \text{The total traffic stored in queue } c \text{ at slot } t \text{ in satellite } n \\ \text{Data packets stored in queue } c \text{ for } l \text{ time slots} \\ \text{The illumination status of cell } c \text{ by satellite } n \text{ in time slot } t \\ L_{n}^{t} \end{array} $
L_n^t The traffic load of satellite n at slot t
L_n^t The traffic load of satellite n at slot t
L_n^t The traffic load of satellite n at slot t
$\tau_{n,c}^{t}$ Average queue delay per data packet in cell c at time slot t
T_{rx} The noise temperature of the receiver
P_{tot} The total power of each satellite
$p_{n,k}^t$ Power of beam k belonging to satellite n
$P_{max}^{n,n}$ The maximum power allocated to each beam
$h_{k_c,l}$ Channel coefficients from beam l to user k_c in cell c.
$G_t(\theta_{k_c,l})$ Transmit antenna gain from the <i>l</i> -th beam to the k_c -th user
$G_r^{k_c}$ Receive antenna gain of the user k_c
$r_{n,c}^t$ Channel capacity of cell c served by satellite n in slot t
$\omega_{i,j}$ The distance between cells <i>i</i> and <i>j</i>
$\overline{\omega}$ The minimum interference distance between cells
$Th_{n,c}^t$ The throughput of cell c during time slot t

A. System Model

1) Digital Twin Network: This paper considers the forward link of a multi-beam LEO satellite communication system in a DT-empowered satellite network, consisting of ground users, LEO satellites, and a cloud center. As shown in Fig. 1, LEO satellites serve users in ground coverage cells through BH. Adjacent LEO satellites have overlapping coverage areas, meaning that multiple satellites may cover the same cell at the same time.

Adjacent LEO satellites have overlapping coverage areas, meaning multiple satellites may cover a cell. The cloud center maintains a DT model of the entire network to monitor and



Fig. 1: Forward link of the DT-empowered multi-beam LEO satellite communication system.

optimize resource allocation. It periodically collects global network state information, such as cell requests, LEO satellite resource usage, and buffer queue backlogs. Using this DT, the cloud center monitors dynamic changes in the LEO satellite network and optimizes BH plans. The DT network determines the service cells for each time slot, encompassing the allocation of overlapping cells.

Consider that there are N satellites serving the ground area containing a total of M cells. The sets of satellites and cells are denoted by $\mathcal{N} = \{n | n = 1, 2, \dots, N\}$, and $\mathcal{V} = \{v | v = 1, 2, \dots, V\}$, respectively. Each satellite can produce K beams and provide transmission service to C cells through timedivision multiplexing (TDM) manner. The set of K beams of satellite n is expressed as $\mathcal{K}_n = \{k_n | k_n = 1, \dots, k, \dots, K\}$. In particular, the set of cells covered by n-th satellite is represented by V_n , $|V_n| = C$. Owing to the multiple coverage characteristics of the LEO satellite communication system [46], one cell may be covered by multiple satellites, i.e. $V_i \cap V_j \neq \emptyset, \forall i, j \in \mathcal{N}$. Without loss of generality, we consider the first T time slots, denoted as $\mathcal{T} = \{t | t = 1, \dots, t, \dots, T\}$. The duration of each time slot is T_{slot} .

2) Traffic Model: Assuming that each satellite holds C queues for storing the arriving traffic from the coverage cells, and each queue capable of storing only the traffic that arrived within the last T_{ttl} time slots due to the limited queue storage, where T_{ttl} is defined as Time To Live (TTL) [41]. $\Lambda_n^t = \{\rho_{n,1}^t, \cdots, \rho_{n,c}^t, \cdots, \rho_{n,C}^t\}$ denotes the arrival traffic of satellite n in time slot t, where $\rho_{n,c}^t$ represents the arrival traffic of cell c covered by satellite n in time slot t. Therefore, the total traffic stored in satellite n during time slot t is represented as $D_n^t = \{d_{n,1}^t, \cdots, d_{n,c}^t, \cdots, d_{n,C}^t\}$, where $d_{n,c}^t$ is the traffic demand of cell c in time slot t. We can reasonably expand $d_{n,c}^t$ to $\{\phi_{c,1}^{n,t}, \phi_{c,2}^{n,t}, \cdots, \phi_{c,T_{ttl}}^{n,t}\}$, where $\sum_{l=1}^{T_{ttl}} \phi_{c,l}^{n,t} = d_{n,c}^t$ and $\phi_{c,l}^{n,t}$ represents the number of data packets that have been in queue c for l time slots. For users in overlapping coverage cells, all covering satellites will receive their requests.

To satisfy the traffic demand of cells, each satellite should reasonably decide the beam irradiation positions. The BH pattern of satellite n at time slot t can be expressed as

$$\boldsymbol{X}_{n}^{t} = \{ x_{n,1}^{t}, \dots, x_{n,c}^{t}, \dots, x_{n,C}^{t} \}$$
(1)

where $x_{n,c}^t \in \{0,1\}$ denotes whether the coverage cell c of satellite n is illuminated by the beam of satellite n in time slot t.

3) Communication Model: In order to fully utilize spectrum resources and enhance spectrum efficiency, all beams are considered to occupy the entire satellite bandwidth B. This implies the use of full frequency reuse in the multisatellite BH system, where co-frequency interference between beams becomes significant. For example, a user in the cell cserved by the k-th beam (referred to as the k_c -th user) will receive NK signals from N satellites. The channel vector linking the user k_c to NK beams is expressed as h_{k_c} = $[h_{k_c,1}, h_{k_c,2}, \ldots, h_{k_c,NK}]$. Assuming the user compensates for the Doppler frequency offset due to satellite motion and that clear sky conditions prevail, neglecting rain attenuation. Let $G_t(\theta_{k_c,l})$ denote the transmit antenna gain of the *l*-th beam to user k_c , which is determined by the off-axis angle $\theta_{k_c,l}$ between user k_c and the main lobe direction of beam l. The receiving antenna gain for user k_c is denoted by $G_r^{k_c}$. So, the channel coefficient between the k_c -th user and the l-th beam can be denoted by [39],

$$h_{k_c,l} = \frac{\sqrt{G_t\left(\theta_{k_c,l}\right)G_r^{k_c}}}{4\pi\frac{d_{k_c,l}}{\lambda'}} \tag{2}$$

where $d_{k_c,l}$ is the distance between user k_c and beam l, and λ' represents the wavelength.

Therefore, the signal-to-noise ratio of user k_c can be expressed as,

$$\operatorname{SINR}_{n,k_c}^{t} = \frac{p_{n,k}^{t} \left| h_{k_c,k} \right|^2}{\sigma^2 + I_{Intra} + I_{Inter}} \tag{3}$$

where $\sigma^2 = k_B T_{rx} B$ represents the noise power. k_B is the Boltzmann constant and T_{rx} is the receiver noise temperature. $I_{Intra} = \sum_{l \in \mathcal{K}_n \setminus \{k\}} p_{n,l}^t |h_{k_c,l}|^2$ denotes intra-satellite interference, $I_{Inter} = \sum_{n' \in \mathcal{N} \setminus \{n\}} \sum_{l \in \mathcal{K}_{n'}} p_{n',l}^t |h_{k_c,l}|^2$ represents inter-satellite interference. $p_{n,l}^t$ is the power allocated by satellite *n* to its *l*-th beam. Similarly, $p_{n',l}^t$ is the power allocated by satellite *n'* to its *l*-th beam.

According to (3), let $r_{n,c}^t$ denote the channel capacity of cell c served by satellite n in time slot t which can be calculated as follows:

$$r_{n,c}^{t} = x_{n,c}^{t} B \log_2\left(1 + SINR_{n,k_c}^{t}\right) \tag{4}$$

where $x_{n,c}^t$ denotes if the cell c is illuminated, and B denotes the bandwidth allocated to beam k serving cell c. Since the system employs full-frequency multiplexing, each beam utilizes the entire available bandwidth B.

Let $Th_{n,c}^t$ denote throughput of cell c at time slot t, which can be calculated as follows:

$$Th_{n,c}^{t} = \min\left\{r_{n,c}^{t}, d_{n,c}^{t}\right\}$$
(5)

B. Optimization Problem Formulation

It should be noted that multiple beams from different satellites may serve the same cell at the same time slot, resulting in significant inter-satellite interference. Therefore, it is necessary to carefully plan the BH pattern of each satellite and the corresponding PA of the beams. Given the timevarying characteristics of traffic demands, the joint BH and PA aims to maximize the total throughput of the satellite while minimizing delay fairness among all cells, given the limitations of total bandwidth and power resources.

The total throughput of all satellites can be expressed as follows:

$$Th_{tot} = \sum_{t=1}^{T} \sum_{n=1}^{N} \sum_{c=1}^{C} Th_{n,c}^{t}$$
(6)

Likewise, let $\tau_{n,c}^t$ denote the average queueing delay of each data packet of the cell c during time slot t, which can be computed by

$$\tau_{n,c}^{t} = \frac{\sum_{l=1}^{T_{ttl}} l \cdot \phi_{c,l}^{n,t}}{\sum_{l=1}^{T_{ttl}} \phi_{c,l}^{n,t}}$$
(7)

Therefore, the delay fairness among all cells J is expressed as follows:

$$J = \sum_{n=1}^{N} \sum_{t=1}^{T} \left(\max_{c \in V_n} \{ \tau_{n,c}^t \} - \min_{c \in V_n} \{ \tau_{n,c}^t \} \right)$$
(8)

To maximize throughput while reducing latency fairness, we establish a optimization problem. This problem combines BH and PA, and can be modeled as follows:

$$P_{0}: \max_{\boldsymbol{X}_{n}^{t}, \boldsymbol{P}_{n}^{t}} \beta \frac{Th_{tot}}{Th_{norm}} - (1-\beta) \frac{J}{J_{norm}}, \beta \in [0,1]$$

s.t. $C_{1}: x_{n,c}^{t} \in \{0,1\}, \forall n \in \mathcal{N}, \forall c \in V_{n}$
 $C_{2}: \sum_{c=1}^{C} x_{n,c}^{t} = K, \forall n \in \mathcal{N}$
 $C_{3}: \sum_{n} x_{n,c}^{t} \leq 1, \forall n \in \mathcal{N}$
 $C_{4}: x_{n,c}^{t} = 0, \forall c \notin V_{n}, \forall n \in \mathcal{N}$
 $C_{5}: x_{n,i}^{t} x_{n',j}^{t} = 1, \omega_{i,j} \geq \varpi, \forall i \in V_{n}, j \in V_{n'}$
 $C_{6}: p_{n,k}^{t} \leq P_{max}, \forall n \in \mathcal{N}, \forall k \in \mathcal{K}_{n}$
 $C_{7}: \sum_{k=1}^{K} p_{n,k}^{t} \leq P_{tot}, \forall n \in \mathcal{N}$

$$(9)$$

In the aforementioned optimization problem, X_n^t and P_n^t are the variables to be optimized, representing the BH pattern and the corresponding beam PA of satellite n at time slot t, respectively. The parameter β is a preset weight used to balance the trade-off between throughput and delay fairness, while Th_{max} and J_{max} are normalization constants. C_1 constrains optimization variables $x_{n,c}^t$ to be binary variables. C_2 states that each satellite can activate exactly K beams at once, meaning each satellite can select K cells within its coverage area for communication services in each time slot. C_3 emphasizes that each cell is served by at most one beam from one satellite in each time slot. C_4 means that each satellite can only select cells within its coverage area for service. C_5 enforces that the distance between cells i and j serviced in time slot t exceeds the minimum interference distance ϖ , with $\omega_{i,j} = \operatorname{dist}(i,j)$ denoting the distance between cells i and j. C_6 limits the power $p_{n,k}^t$ allocated to each beam not exceeding the maximum beam power P_{max} . C_7 ensures that the total beam power of satellite n does not exceed the total power of satellite P_{tot} .

It can be observed that problem (9) is both non-convex and nonlinear, with the non-convexity arising from the SINR in equation (3). Additionally, problem (9) involves continuous variables $p_{n,k}^t$ and discrete variables $x_{n,c}^t$. As a result, this optimization problem is NP-hard, making it impractical to find an optimal solution in polynomial time. In the following section, we will introduce a new method that provides a suboptimal solution with low computational complexity.

IV. PROBLEM DECOMPOSITION AND PROPOSED BHPA-LBDP ALGORITHM

In this section, P_0 is divided into two sub-problems: the BH design problem for multiple satellites based on load balancing and interference avoidance, and the PA problem for each satellite.

A. Multi-Satellite BH Design Based on Load Balancing and Interference Avoidance

In LEO satellite BH systems, the distribution of ground services is uneven, resulting in significant differences in service demands across satellites. Certain satellites cover highdemand, densely populated regions, whereas others cover sparsely populated areas with relatively low demand. If all high-demand cells were assigned to a single satellite, it would become overloaded, leaving the resources of other satellites underutilized. As a result, balancing the load among satellites is crucial for improving the system's overall resource efficiency.

Since hotspot areas can be covered by multiple neighboring satellites simultaneously, serving the same hotspot cell with multiple satellites at the same time can result in severe inter-satellite interference. Therefore, assigning these hotspot areas to different satellites can reduce the load on individual satellites and avoid interference among neighboring satellites.

1) Optimization Problem Formulation: In summary, the multi-satellite traffic-driven BH problem can be described as minimizing the weighted sum of traffic load between satellites and cell service fairness, which is formalized as follow:

$$P_{1}:\min_{\boldsymbol{X}_{n}^{t}} \left\{ \alpha \frac{Q}{Q_{max}} + (1-\alpha) \frac{J}{J_{max}} \right\}$$

s.t. $C_{1}: Q = \sum_{t=1}^{T} \max_{n \in \mathcal{N}} \left\{ L_{n}^{t} \right\} - \sum_{t=1}^{T} \min_{n \in \mathcal{N}} \left\{ L_{n}^{t} \right\}$
 $C_{2}: L_{n}^{t} = \sum_{c \in V_{n}} d_{n,c}^{t} x_{n,c}^{t}, \forall n \in \mathcal{N}$
 $C_{3}: C_{1} - C_{5} \quad \text{from} \quad P_{0}$ (10)

where X_n^t is the only optimization variable of the optimization problem, representing the BH pattern of satellite n at time slot t. In C_1 , Q represents the load balance among satellites, defined as the difference between the maximum and minimum satellite loads. L_n^t represents the traffic load of satellite n, which is calculated in C_2 .

It should be noted that the optimization problem (10) is an integer, non-linear, and non-convex problem, which is still NP-hard. Additionally, due to the rapid movement of LEO satellites, the coverage cells of the satellites are dynamically changing, that is, the arrival traffic of the coverage cells received by the satellites is frequently changing. Thus, traditional integer optimization methods, such as branch and bound, cannot be applied here. In this paper, an DRL algorithm is proposed to find the relatively optimal BH decision through continuously interacting with the environment.

2) Multi-Agent MDP Formulation: In fact, the BH optimization problem (10) is essentially a sequential decisionmaking problem, and the traffic D^t can be expressed as,

$$D^{t} = D^{t-1} - X^{t-1}r^{t-1} + \Lambda^{t}$$
(11)

However, the request arrival traffic state Λ^t is unknown during BH decision-making. Fortunately, the states from previous time slots are easily accessible. Given the temporal correlation between state changes, we are motivated to predict the current state by leveraging past state sequences.

Arrival Traffic Estimation: The LSTM model is trained using historical request arrival traffic state sequences of cells, which are collected by DT. The error between the actual state and the predicted state is used to update the weight parameters W and U. At time step t - 1, the request arrival traffic state $\rho_{n,c}^{t-1}$ serves as the input to the LSTM network, with the output providing an estimate of the current request arrival traffic state. The memory block parameters are updated as follows:

$$f_{t} = \sigma \left[W_{f} \rho_{n,c}^{t-1} + U_{f} h_{t-1} + b_{f} \right]$$

$$i_{t} = \sigma \left[W_{i} \rho_{n,c}^{t-1} + U_{i} h_{t-1} + b_{i} \right]$$

$$\hat{C}_{t} = \tanh \left[W_{c} \rho_{n,c}^{t-1} + U_{c} h_{t-1} + b_{c} \right]$$

$$C_{t} = f_{t} \odot C_{t-1} + i_{t} \odot \widetilde{C}_{t}$$

$$o_{t} = \sigma \left[W_{o} \rho_{n,c}^{t-1} + U_{o} h_{t-1} + b_{o} \right]$$

$$h_{t} = o_{t} \odot \tanh (C_{t})$$

$$\hat{o}_{n,c}^{t} = W_{h} h_{t} + b_{h}$$
(12)

Here, W_f represents the weight between the input and the forget gate, while U_f corresponds to the weight between the previous hidden state h_{t-1} and the forget gate. The bias is b_f , and $\sigma(\cdot)$ denotes logical sigmoid function. W_i and W_c are the weights linking the input state $\rho_{n,c}^{t-1}$ to the input gates, whereas U_i and U_c represent the weights between the previous hidden state ht - 1 and the input gate. The hyperbolic tangent function is $\tanh(\cdot)$, and \odot stands for the Hadamard product. W_o refers to the weight between the current input $\rho_{n,c}^{t-1}$ and the output gate, and U_f represents the weight between the hidden state h_{t-1} and the output gate. Finally, W_h and b_h are the output weight matrix and bias term corresponding to the predicted state $\hat{\rho}_{n,c}^t$, respectively.

Í

Thus, we can model the state evolution using a Markov Decision Process (MDP). Specifically, the four key components of the MDP, i.e., state, observation, action, and reward, are defined as follows. • **State:** At time slot *t*, the system can observe the entire traffic demand as the global state, which is represented as follows:

$$\boldsymbol{s}^{t} = [\operatorname{vec}(\hat{\mathbf{D}}^{t}); \operatorname{vec}(\mathbf{H}^{t})]$$

$$\hat{\boldsymbol{D}}^{t} = \boldsymbol{D}^{t-1} - \boldsymbol{X}^{t-1} \boldsymbol{r}^{t-1} + \hat{\boldsymbol{\Lambda}}^{t}$$

$$\boldsymbol{D}^{t-1} = \begin{bmatrix} d_{1,1}^{t-1} & d_{1,2}^{t-1} & \cdots & d_{1,C}^{t-1} \\ d_{2,1}^{t-1} & d_{2,2}^{t-1} & \cdots & d_{2,C}^{t-1} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ d_{N,1}^{t-1} & d_{N,2}^{t-1} & \cdots & d_{N,C}^{t-1} \end{bmatrix}$$

$$\hat{\boldsymbol{\Lambda}}^{t} = \begin{bmatrix} \hat{\rho}_{1,1}^{t} & \hat{\rho}_{1,2}^{t} & \cdots & \hat{\rho}_{1,C}^{t} \\ \hat{\rho}_{2,1}^{t} & \hat{\rho}_{2,2}^{t} & \cdots & \hat{\rho}_{2,C}^{t} \\ \vdots & \vdots & \vdots & \vdots \\ \hat{\boldsymbol{\rho}}_{N,1}^{t} & \hat{\rho}_{N,2}^{t} & \cdots & \hat{\rho}_{N,C}^{t} \end{bmatrix}$$

$$\boldsymbol{H}^{t} = \begin{bmatrix} h_{1,1}^{t} & h_{1,2}^{t} & \cdots & h_{1,C}^{t} \\ h_{2,1}^{t} & h_{2,2}^{t} & \cdots & h_{2,C}^{t} \\ \vdots & \vdots & \vdots \\ h_{N,1}^{t} & h_{N,2}^{t} & \cdots & h_{N,C}^{t} \end{bmatrix}$$
(13)

where $\hat{\rho}_{n,c}^t$ represents the predicted request arrival traffic in cell c covered by satellite n at time slot t, which can be obtained by (12). $h_{n,k}^t$ is channel gain from satellite n to its c-th cell at time slot t, which can be obtained by (2). It should be noted that the state represents global information, which the agents (i.e., the satellites) cannot fully access since they only have local observations. Therefore, the global state can only be utilized by the centralized trainer in DT network.

• Observation: Denote o_n^t as the local observation accessed by the *n*-th agent at time slot *t*. Similar to the state, the local observation,

$$\boldsymbol{o}_n^t = [\boldsymbol{D}_n^t; \boldsymbol{H}_n^t] \tag{14}$$

where $\boldsymbol{D}_{n}^{t} = \begin{bmatrix} d_{n,1}^{t}, d_{n,2}^{t}, \dots, d_{n,C}^{t} \end{bmatrix}$ represents the status of traffic queues on the *n*-th satellite. $\boldsymbol{H}_{n}^{t} = \begin{bmatrix} h_{n,1}^{t}, h_{n,2}^{t}, \dots, h_{n,C}^{t} \end{bmatrix}$ is the channel information of the *n*-th satellite.

• Action: As an agent, the satellite must decide which cells to illuminate. Thus, the action of agent can be defined as:

$$a_{n}^{t}\big|_{n \in N} = \left(x_{n,1}^{t}, \dots, x_{n,c}^{t}, \dots, x_{n,C}^{t}\right),$$

$$x_{n,c}^{t} \in \{0,1\}, \text{and} \sum_{c=1}^{C} x_{n,c}^{t} = 1$$
(15)

• **Reward:** To minimize the objective defined in equation (10), we design the reward at time t as follows:

$$R^{t} = R^{t} \left(s^{t}, a^{t} \right) = -\left[\alpha \frac{\max \left\{ L_{n}^{t} \right\} - \min \left\{ L_{n}^{t} \right\}}{Q_{\max}} + (1 - \alpha) \frac{\max \left\{ \tau_{n,i}^{t} \right\} - \min \left\{ \tau_{n,j}^{t} \right\}}{J_{\max}} + \Gamma \right], \forall i, j \in V_{n}$$
(16)

Note that we introduce a penalty term Γ in the reward to avoid inter-satellite interference. This penalty is proportional to the number of cells illuminated simultaneously. Thus, the reward will be reduced if multiple satellites illuminate the same cell at a time slot.

The request arrival traffic estimation module deployed in DT can predict the state $\hat{\rho}_n^t$ for time slot t based on the observed state ρ_n^{t-1} for time slot t-1. The predicted state, along with the local queue state and channel state of time slot t is then input into the local cooperative BH strategy module. This module outputs the local BH action a that the agent should take at decision time t, thereby determining which cells will be served by satellite n.

3) Multi-Agent BH Decision Based on A3C (MA3C-BH): Since the Actor-Critic method performs better in discrete action spaces, this paper employs a multi-agent A3C algorithm based on Actor-Critic method.



Fig. 2: The cooperative multi-agent actor-critic framework of the MA3C-BH algorithm.

Note that the BH decision is made in the DT, which has sufficient computing power. As shown in Fig.2, each satellite deploys an actor network to select actions and a critic network to estimate the value of actions. Simultaneously, the DT deploys a critic network to integrate the feedback of all agents and provide global value evaluation.

This study establishes I + 1 threads based on the A3C algorithm. Each thread designs N actor networks and N + 1critic networks as illustrated in Fig. 2, where N represents the total number of satellites. The network parameters in each thread are denoted as $\{\theta_1, \theta_2, \cdots, \theta_N, \phi_1, \phi_2, \cdots, \phi_N\}$. The network $\{\theta_1, \theta_2, \cdots, \theta_N, \phi_1, \phi_2, \cdots, \phi_N\}$ outputs a decision network $\{a_1^t, a_2^t, \cdots, a_N^t\}$ based on the observation vector $\{\boldsymbol{o}_1^t, \boldsymbol{o}_2^t, \cdots, \boldsymbol{o}_N^t\}$ obtained from the environment. The Critic network $\{\phi_1, \phi_2, \cdots, \phi_N\}$ calculates the Temporal Difference-error (TD-error) δ based on the reward r^t generated at that moment and passes it to the other actor networks. Each actor network uses the received TD-error δ to update its parameters via gradient descent.

In the A3C architecture, the global network does not participate in actual exploration. Instead, it copies and shares the parameters explored by the preceding threads, achieving asynchronous learning and accelerating convergence. In a separate

Algorithm 1 Multi-agent A3C BH strategy (MA3C-BH)

- 1: Initialize global parameters θ_g for actor network, ϕ_g for critic network
- 2: Initialize thread-specific parameters θ^i for actor network, ϕ^i for critic network
- 3: Initialize shared critic network parameters ϕ_s
- 4: Set the number of threads I and the number of actor-critic networks per thread N
- 5: for each thread $i = 1, \ldots, I$ do
- Initialize environment 6:
- Initialize local copy of parameters $\theta^i \leftarrow \theta_q, \ \phi^i \leftarrow \phi_q$ 7:
- while not converged do 8:
- Receive state s^t from environment 9:
- for each actor-critic network $j = 1, \ldots, N$ do 10:
- Calculate action $a_j^t = \pi_{\theta_i^i}(s^t)$ using actor 11: network
- Execute action a_i^t and observe reward r_i^t and 12: next state s_j^{t+1}
- Calculate TD-error $\delta_i^i = r_i^t + \gamma V_{\phi_i^i}(s_i^{t+1}) -$ 13: $V_{\phi^i}(s^t)$
- Update actor network parameters θ_i^i using pol-14: icy gradient by: $\Delta \theta_j^i = \nabla_{\theta_j^i} \log \pi_{\theta_j^i} (o_j^t, a_j^t) \delta_j^{i'}$ Update local Critic network parameters ϕ_j^i us-
- 15: ing TD-error: $\phi_j^i \leftarrow \phi_j^i + \beta \delta_j^i \nabla_{\phi_a^i} V_{\phi_a^i}(s^t)$
- end for 16:
- Update shared critic network parameters ϕ_s using 17: average TD error by (19)
- 18: end while
- end for 19:
- 20: Asynchronously update global parameters:

$$\theta_g \leftarrow \frac{1}{I} \sum_{i=1}^{I} \theta^i$$
$$\phi_g \leftarrow \frac{1}{I} \sum_{i=1}^{I} \phi^i$$

thread, the critic network predicts the value function at two moments through the states X(t) and X(t+1), calculates the TD-error, and updates the network parameters by minimizing the least squares temporal difference (LSTD),

$$V^* = \arg\min_{V} \left(\delta^i_{\theta_c}\right)^2 \tag{17}$$

where V^* represents the optimization value function. For the actor network, after receiving the TD-error calculated by the critic network, the TD-error is used to calculate its gradient:

$$\nabla_{\theta_{j}^{i}}J\left(\theta_{j}^{i}\right) = \mathbb{E}\left[\nabla_{\theta_{j}^{i}}\log\pi_{\theta_{j}^{i}}\left(o_{j}^{t},a_{j}^{t}\right)\delta_{j}^{i}\right]$$
(18)

where $\pi_{\theta_{i}^{i}}(s, a)$ represents the probability of the Actor network taking action a_i^t under the state observation vector o_i^t .

We then use gradient descent to update the actor network:

$$\theta_j^i \leftarrow \theta_j^i + \alpha \nabla_{\theta_j^i} \log \pi_{\theta_j^i} \left(o_j^t, a_j^t \right) \delta_j^i \tag{19}$$

Additionally, each agent will also use the shared critic network to update its value function:

$$\phi_s \leftarrow \phi_s + \lambda \frac{1}{V} \sum_{j=1}^{V} \delta_j^i \nabla_{\phi_s} V_{\phi_s}(s^t)$$
(20)

In summary, the proposed algorithm MA3C-BH is showing in Algorithm 1.

B. PA Based on Delay Fairness

After solving problem P_1 , the BH pattern is determined, and each satellite has to determine the PA. This scheme should promote the transmission of a larger proportion of packets in the satellite's buffer queue, preventing packet loss due to queuing delays exceeding the maximum tolerable threshold. This strategy is meticulously designed to enhance the satellite's data throughput while ensuring minimal delay fairness among the served cells, thus balancing efficiency and equity in resource allocation.

1) Problem Formulation : We define $p_{n,k}^t$ to represent the power allocated to the beam k of satellite n during slot t. From problem P_0 we can formulate the PA problem as:

$$P_{2}: \max_{\boldsymbol{P}_{n}^{t}} \left\{ \beta \frac{Th_{tot}}{Th_{max}} - (1-\beta) \frac{J}{J_{max}} \right\}, \beta \in [0,1]$$

s.t. $C_{1}: C_{6}, C_{7}$ from P_{0} (21)

where $\boldsymbol{P}_{n}^{t} = \{p_{n,1}^{t}, \dots, p_{n,k}^{t}, \dots, p_{n,K}^{t}\}$ represents the power allocated to each beam of satellite n.

2) The Sequential Decision-Making Problem Reformulation: The problem P_2 can be regarded as a sequence decisionmaking problem, which can be further reformulated as an MDP with infinite states and continuous action spaces. Define the formulated MDP as $(s^{t-1}, a^{t-1}, r^{t-1}, s^t)$, where s represents the state set of the environment, a indicates the action taken by the agent, and r refers to the reward function.

• State: The states of systems consists of two parts: the matrix of the number of packets \overline{D}_n^t and the downlink loss matrix \overline{H}_n^t . Thus, the state vector s^t can be defined as:

$$\boldsymbol{s}^{t} = (\overline{\boldsymbol{D}}_{n}^{t}, \overline{\boldsymbol{H}}_{n}^{t}),$$

$$\overline{\boldsymbol{D}}_{n}^{t} = [d_{n,c}^{t} | \boldsymbol{x}_{n,c}^{t} = 1, \forall c \in V_{n}]$$

$$= [d_{n,1}^{t}, \dots, d_{n,k}^{t}, \dots, d_{n,K}^{t}]$$

$$\overline{\boldsymbol{H}}_{n}^{t} = [h_{n,c}^{t} | \boldsymbol{x}_{n,c}^{t} = 1, \forall c \in V_{n}]$$

$$= [h_{n,1}^{t}, \dots, h_{n,k}^{t}, \dots, h_{n,K}^{t}]$$
(22)

where \overline{D}_n^t represents the traffic of K cells selected to be served by satellite n in time slot t, with $d_{n,k}^t$ and $h_{n,k}^t$ representing the traffic and channel gain of the selected k-th cell, respectively.

• Action: In the LEO layer, each beam is considered as an agent that decides PA in each time slot. Thus, the actions can be defined as:

$$\boldsymbol{a}^{t} = \boldsymbol{a}_{n}^{t} = \begin{bmatrix} p_{n,1}^{t}, \dots, p_{n,k}^{t}, \dots, p_{n,K}^{t} \end{bmatrix}$$
(23)

where $p_{n,k}^t$ denotes the power that the satellite *n* allocates to the beam *k* at time slot *t*.

• Reward: To promote cooperation among the agents, they are awarded a global reward, which is defined as follows:

$$R^{t} = R^{t} \left(s_{n}^{t}, a_{n}^{t}\right) = \beta \frac{\sum_{c=1}^{C} Th_{n,c}^{t}}{Th_{\max}} - (1-\beta) \frac{\left(\max\left\{\tau_{n,i}^{t}\right\} - \min\left\{\tau_{n,j}^{t}\right\}\right)}{J_{\max}}, \forall i, j \in V_{n}$$

3) PA Based on MADDPG (MAPA): We employ the MAD-DPG algorithm, a multi-agent deep reinforcement learning method that merges the Actor-Critic (AC) and Deep Deterministic Policy Gradient (DDPG) algorithms, to solve the reformulated MDP problem. Since all onboard transmitters are deployed on the same satellite, each beam can observe the same state information. The decision for each beam is made by its own policy network, independent of other beams' actions. This process is known as decentralized decision-making. As shown in Fig.3, an actor network is deployed by each beam to choose actions, and a global critic network is used by the satellite to assess the action values.



Fig. 3: The cooperative multi-agent actor-critic framework of the MAPA algorithm.

However, due to the cooperative and competitive relationships among the beams, an individual beam cannot reach an optimal solution without considering the actions of other beams. Therefore, after the beams make decentralized decisions, a policy evaluation network assesses the overall strategy of all beams on the satellite. The evaluation results are fed back to each beam's policy network to update the strategy parameters. Through continuous learning and iteration, the multiple beams can ultimately achieve a unique optimal solution. This process is known as centralized training.

In this work, each beam is treated as an agent and is equipped with its own actor network. The task of the actor network is to select actions, represented as:

$$\boldsymbol{a}_{i}^{t} = \pi_{\theta_{i}^{t}}(s_{i}^{t}) + \mathcal{N}_{0} \tag{25}$$

where \mathcal{N}_0 represents exploration noise added to promote diverse action exploration. Concurrently, all agents execute their chosen actions a_i , and the environment responds with

a new state s'_i and an immediate reward r_i . The local critic network then calculates the Q value, y_i , as

$$y_i^t = r_i^t + \gamma Q_{\phi'}(\{s_i^{t+1}\}, \{a_i^{t+1}\})|_{a_j^{t+1} = \pi_{\theta_{j'}}(s_j^{t+1})}$$
(26)

Updates the global critic network parameters by minimizing the Q value error:

$$L(\phi) = \frac{1}{|R|} \sum_{j} \left(y_j - Q_\phi \left(\{s_j\}, \{a_j\} \right) \right)^2$$
(27)

At the same time, each agent's local actor network parameters are updated using the policy gradient:

$$\nabla_{\theta^{i}} J \approx \frac{1}{|R|} \sum_{j} \nabla_{a_{i}} Q_{\phi}(\{s_{j}\}, \{a_{j}\})|_{a_{i}=\pi_{\theta^{i}}(s_{i})} \nabla_{\theta^{i}} \pi_{\theta^{i}}(s_{i})$$

$$(28)$$

The specific procedure of the algorithm, called MAPA, is summarized in Algorithm 2.

Algorithm 2 MADDPG for PA (MAPA)

- 1: Initialize environment with satellites, beams, and cells, global power constraint P
- 2: Initialize actor network parameters θ_i for each agent i
- 3: Initialize global critic network parameters ϕ
- 4: Initialize target network parameters $\theta_{i'} \leftarrow \theta_i, \phi' \leftarrow \phi$
- 5: Set learning rates $\alpha_{\theta}, \alpha_{\phi}$
- 6: Initialize replay buffer \mathcal{R}
- 7: Set the number of beams K, the number of time slots T
- 8: Set soft update parameter τ
- 9: for each episode do
- 10: Initialize state s_i^0 for each agent *i*
- 11: **for** each time slot $t = 0, \ldots, T$ **do**
- 12: for each agent $i = 1, \ldots, K$ do
- 13: Select action a_i^t according to (25)
- 14: end for
- 15: Execute actions $\{a_i^t\}$ and observe new states $\{s_i^{t+1}\}$ and rewards $\{r_i^t\}$
- 16: Store transition $(\{s_i^t\}, \{a_i^t\}, \{r_i^t\}, \{s_i^{t+1}\})$ in replay buffer R
- 17: Sample a random minibatch of transitions $(\{s_i^t\}, \{a_i^t\}, \{r_i^t\}, \{s_i^{t+1}\})$ from \mathcal{R}

```
Global critic network update:
18:
             for each agent i do
19:
                 Set (26)
20:
             end for
21:
             Update critic by minimizing the loss: (27)
22.
             actor network update:
23:
             for each agent i do
24:
25:
                 Update actor using sampled policy gradient:
                 (28)
26:
             end for
27:
             Soft update target networks:
28:
            \theta_{i'} \leftarrow \tau \theta_i + (1 - \tau) \theta_{i'}; \phi' \leftarrow \tau \phi + (1 - \tau) \phi
```

29: end for

30: end for

C. Complexity Analysis

First, we analyze the complexity of using LSTM to predict the demand of a community. Assume that the LSTM network has L layers, each layer has H hidden units, and the time step is T. Then the complexity of LSTM is $\mathcal{O}(L_{lstm} \cdot H_{lstm}^2 \cdot T)$.

Secondly, we assess the complexity involved in solving sub-problem P_1 . In the A3C algorithm adopted, each agent (satellite) has an Actor-Critic network, and there is a Critic network is centrally deployed. Assuming that the Actor network has L_{θ} layers, each layer has H_{θ} neurons, its complexity is $\mathcal{O}(L_{\theta} \cdot H_{\theta}^2)$. For the Critic network, assuming that there are L_{ϕ} layers, and each layer has H_{ϕ} neurons, the complexity of the local and global Critic network is: $\mathcal{O}(L_{\phi} \cdot H_{\phi}^2)$. Thus, the total complexity at each time step can be expressed as $\mathcal{O}(N \cdot (L_{\theta} \cdot H_{\theta}^2 + 2L_{\phi} \cdot H_{\phi}^2)) = \mathcal{O}(N \cdot (L_{\theta} \cdot H_{\theta}^2 + L_{\phi} \cdot H_{\phi}^2))$. In addition, the complexity of asynchronous global parameter update is $\mathcal{O}(I \cdot (L_{\theta} \cdot H_{\theta}^2 + L_{\phi} \cdot H_{\phi}^2))$. As a result, the total complexity for solving sub-problem P_1 is $\mathcal{O}(I \cdot T \cdot N \cdot (L_{\theta} \cdot H_{\theta}^2 + L_{\phi} \cdot H_{\phi}^2))$.

Thirdly, the complexity of the MAPA algorithm for addressing sub-problem P_2 (i.e., the single-satellite PA problem) is analyzed. Similarly, assuming that the Actor network for each beam has $L_{\theta'}$ layers, each layer has $H_{\theta'}$ hidden neurons, and the input dimension is D_A , the Actor network's complexity is expressed as $\mathcal{O}(\mathcal{B} \cdot L_{\theta'} \cdot H_{\theta'}^2)$. For the Critic network, assuming that the Critic network has $L_{\phi'}$ layers, and each layer has $H_{\phi'}$ neurons, the Critic network's complexity is given by $\mathcal{O}(\mathcal{B} \cdot L_{\phi'} \cdot H_{\phi'}^2)$. Thus, the total complexity of sub-problem P_2 is $\mathcal{O}(T \cdot K \cdot \mathcal{B} \cdot (L_{\theta'} \cdot H_{\theta'}^2 + L_{\phi'} \cdot H_{\phi'}^2)$.

To summarize, the computational complexity of entire algorithm BHPA-LBDP is $\mathcal{O}(L_{lstm} \cdot H_{lstm}^2 \cdot T) + \mathcal{O}(I \cdot T \cdot N \cdot (L_{\theta} \cdot H_{\theta}^2 + L_{\phi} \cdot H_{\phi}^2) + \mathcal{O}(T \cdot K \cdot \mathcal{B} \cdot (L_{\theta'} \cdot H_{\theta'}^2 + L_{\phi'} \cdot H_{\phi'}^2))$

V. PERFORMANCE EVALUATION

This section first describes the simulation environment and the parameters used for the simulations. Then, we present the simulation results to demonstrate the performance in DTempowered multi-beam satellite network with the proposed algorithm BHPA-LBDP.

A. Simulation Parameter

In this paper, we simulate a Ka-band forward BH and PA system. In both MA3C-BH and MAPA network architectures, the actor network comprises two hidden layers with 128 neurons each, while the critic network consists of two hidden layers with 256 neurons each. Adam is used as the solver for both networks. The constellation parameters and other main simulation parameters are summarized in Table II [45].

We present the optimization effects of the proposed BH and PA Algorithm based on Load Balancing and Delay Fairness (named BHPA-LBDF, including MA3C-BH and MAPA) and compare it with four different BHPA schemes as follows:

1) Random Beam Hopping Fixed Power Allocation (RBH-FP): In RBH-FP, each satellite randomly selects four cells to serve in each time slot, and distributes total power equally among the selected cells.

2) Random Beam Hopping Demand Based Power Allocation (RBH-DP): In RBH-DP, each satellite randomly selects four cells to serve in each time slot and allocates power according to the demand ratio of the selected cells.

3) Beam Hopping Fixed Power Allocation with Load Balancing and Delay Fairness (FPA): In FPA, based on load balancing across all satellites and service fairness within each satellite's coverage, four cells are selected for each satellite in each time slot, with total power divided equally among the selected cells.

4) Beam Hopping and Discretized Power Allocation Based on DT (DPA): In DPA, each satellite acts as an agent, completing BH and PA (discretized processing) to maximize system throughput and delay fairness.

TABLE II: System Parameters

Parameters	Values
SYSTEM PARAMETERS	
Satellite altitude	780 km
Ka-band frequency	20 GHz
Number of satellites, N	12
The number of cells served by per satellite, C	19
Total number of cells covered by all satellites, V	168
The coverage radius of a cell, \vec{R}	39 km
Number of beams per satellite, K	4
The available bandwidth for the system, B	100 MHz
Satellite's total power, P	39 dBW
The maximum power for each beam, P_{max}	30 dBW
TAperture radius of the antenna	0.15 m
The satellite beams' 3 dB beamwidth	3°
Maximum gain of the transmit antenna	35.9 dBi
Receiving antenna gain at the terminal, G_r	0 dBi
Temperature of noise, T_{rx}	300 K
Time slot duration, $T_{\rm slot}$	2 ms
The period of BHTP, T_H	$64 \times T_{\rm slot}$
Weighting factors, α, β	$\alpha = \beta = 0: 0.1: 1$
MA3C-BH TRAINING PARAMETERS	
Training Episode	6000
Actor network learning rate, α_{θ}	0.00001
Critic network learning rate, α_{ϕ}	0.0001
Discount factor, γ	0.99
Number of Threads, I	16
Optimizer	Adam
MA-PA TRAINING PARAMETERS	
Training Episode	6000
Soft Update Parameter, τ	0.001
Replay Buffer size, \mathcal{R}	1000000
Minibatch size, $ \mathcal{B} $	64
Discount factor, γ	0.99
Actor network learning rate α_{θ}	0.00001
Critic network learning rate α_{ϕ}	0.0001
Exploration Noise \mathcal{N}_0	0.2
Optimizer	Adam

B. Convergence Analysis

To illustrate the convergence of the BHPA-LBDF method, we present the episode reward and episode loss from one of the training scenarios with $\alpha = \beta = 0.5$, as depicted in Fig. 4. As shown in Fig. 4(a), the reward increases with the training episodes and converges after 1200 episodes. Fig. 4(b) shows that the episode loss remains stable and converges to 10^{-4} after approximately 4000 episodes. This demonstrates that the convergence of BHPA-LBDF is guaranteed under the parameter settings listed in Table II. Next, we will compare the performance of BHPA-LBDF with the four aforementioned schemes. The performance metrics include throughput, load balance, queueing delay and delay fairness.



Fig. 4: An illustration of (a) episode reward and (b) loss during training $(\alpha = \beta = 0.5)$

C. Performance Analysis



Fig. 5: The performance of (a) traffic load-traffic load gap and (b) delay fairness under different optimization weight α .

1) Performance Under Different Optimization Weights: The optimization weight α reflects the trade-off between load balance Q and delay fairness J. As shown in Fig. 5, the performance of traffic load, traffic load gap and delay fairness changes with α . Both traffic load and delay fairness increase as α rises from 0 to 1 since a larger α focuses more on load balancing. This reduces the load gap, thereby improving delay fairness. A higher delay fairness indicates greater inequity in resource allocation. To strike a balance between load and delay fairness, we use the trained model with $\alpha = 0.5$ for comparison with other approaches.



Fig. 6: The performance of (a) throughput and (b) delay fairness under different optimization weight β .

Similarly, the optimization weight β reflects the trade-off between throughput and delay fairness. As shown in Fig. 6, the performance of both throughput and delay fairness improves as β increases from 0 to 1. To achieve a better balance between throughput and delay fairness, we use the trained model with $\beta = 0.5$ for comparison with other approaches.

2) Performance of Load Balancing: Fig. 7 shows the traffic load of each satellite after BH using different algorithms. The red box represents the BH strategy using MA3C algorithm, which considers load balancing and delay fairness, achieving the smallest load difference of 82.13 Mbps between satellites. Notably, due to uneven traffic distribution, MA3C algorithm improves load balancing by 75.95%, 43.45%, 60.95%, and 73.11% compared to the Greedy BH (G-BH), Random BH (R-BH), Periodic BH (P-BH), and Queue-Length Priority BH (Q-BH) methods, respectively.



Fig. 7: Results of load balancing after beam hopping for different methods (the proposed MA3C-BH with α = 0.5).

3) Performance of Throughput: Fig. 8 shows the throughput versus total traffic demand using different methods, and Fig. 9 shows throughput of 12 satellites among different algorithms. The total traffic demand represents the cumulative demand of all cells. The proposed method, BHPA-LBDP, consistently outperforms other algorithms. This improvement occurs because agents accumulate experience during training, eventually learning to fully utilize the beam's degrees of freedom in time, space, and power. Specifically, compared to RBH-FP, RBH-DP, FPA, and DPA, the throughput of the proposed method improves by 96.7%, 46.2%, 10.2%, and 5.6%, respectively.

As the total traffic demand increases, the throughput of the BHPA-LBDP, FPA, and DPA algorithms consistently surpasses that of the RBH-FP and RBH-DP algorithms. This is mainly because BHPA-LBDP, FPA, and DPA use reinforcement learning to select service cells for BH based on cell demand, whereas RBH-FP and RBH-DP randomly select cells for service. The throughput of RBH-DP is always higher than that of RBH-FP, especially as the total demand increases, with the gap between them gradually widening. This indicates that intelligent PA significantly enhances overall throughput. This is also validated by the comparison of throughput between the BHPA-LBDP and FPA algorithms.

When the total demand is between 8000 and 20000 Mbps, the throughput of the FPA and DPA algorithms shows little difference, with FPA being slightly higher than FPA. This is because both algorithms perform intelligent BH based on cell demand, and the capacity of each beam exceeds the traffic demand of each cell, resulting in a relatively minor impact from PA on overall throughput. However, when the demand exceeds 20000 Mbps, the throughput of DPA becomes significantly higher than that of FPA. This is mainly because, with increasing total demand, DPA can flexibly allocate limited power resources to different cells. Nevertheless, due to the discretization of power by DPA and its joint decision on BHPA within DT, its throughput is always lower than that of our proposed BHPA-LBDP algorithm.



Fig. 8: The comparison of throughput versus total traffic demand for different methods (the proposed BHPA-LBDP with $\alpha = \beta = 0.5$).



Fig. 9: The throughput of 12 satellites among different algorithms. (the proposed BHPA-LBDP with $\alpha = \beta = 0.5$).

4) Performance of Delay Fairness: Fig. 10 illustrates the relationship between the delay fairness of various methods and the total traffic demand. Fig. 11 shows the average queuing delay among different satellites for the five algorithms.

With the increase in total traffic demand, delay fairness improves for all methods, as the limited beam resources are insufficient to handle the excessive traffic, resulting queue congestion. Throughout this process, the queueing delay of intelligent BH algorithms (BHPA-LBDP, FPA, and DPA) remains consistently lower than that of the random BH algorithm. Our proposed BHPA-LBDP algorithm also maintains a lowest queue delay. Compared to the RBH-FP, RBH-DP, FPA, and DPA algorithms, the average queue delay is reduced by 46%, 43.23%, 10.8%, and 20.2%, respectively. Additionally, comparing the three algorithms, i.e., BHPA-LBDP, FPA, and DPA, reveals that PA decision-making based on actual channel information on the satellite has significant advantages in improving overall performance.



Fig. 10: The delay fairness versus total traffic demand of different methods. (the proposed BHPA-LBDP with $\alpha = \beta = 0.5$).



Fig. 11: The delay fairness of 12 satellites among different algorithms. (the proposed BHPA-LBDP with $\alpha = \beta = 0.5$).

VI. CONCLUSION

This paper investigates the coordinated BH and PA problem in multi-beam LEO satellite constellations to align limited satellite resources with uneven service demands. Initially, a digital-twin-based framework for coordinated BH and resource optimization among multiple satellites is established. Acknowledging the NP-hard nature of the optimization task, the problem is decomposed into two subsidiary problems. First, leveraging the multiple coverage offered by LEO satellite constellations, cloud center is employed to select spatially isolated BH patterns for each satellite. This approach balances intersatellite load demands, ensures equitable service provisioning, and mitigates interference. Second, PA is dynamically managed through a collaborative competitive mechanism among beams within each satellite, catering to real-time needs. Finally, simulations validate that this methodology outperforms existing BHPA schemes in terms of thoughput and delay fairness.

In the future, we aim to develop a comprehensive resource allocation strategy that integrates the optimization of BH, spectrum, time, and power resources to maximize long-term user data rates and minimize system power consumption. We will focus on algorithms that leverage the combination of digital twins and artificial intelligence to enable dynamic and intelligent management of onboard resources, ensuring optimal system performance in dynamic environments. Additionally, we plan to introduce intelligent prediction mechanisms to proactively identify and address potential interference and resource bottlenecks, thereby enhancing overall system effi-

resource bottlenecks, thereby enhancing overall system efficiency and reliability. We anticipate that these improvements will significantly improve the effectiveness of our proposed joint BH and multi-dimensional resource allocation framework, creating a more efficient and energy-saving communication environment for LEO satellite networks.

REFERENCES

- [1] J. Sheng, X. Cai, Q. Li, C. Wu, B. Ai, Y. Wang, M. Kadoch, and P. Yu, "Space-air-ground integrated network development and applications in high-speed railways: A survey," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 8, pp. 10066–10085, 2022.
- [2] Q. Ouyang, N. Ye, J. Gao, A. Wang, and L. Zhao, "Joint in-orbit computation and communication for minimizing download time from leo satellites," *IEEE Trans. Mobile Comput.*, vol. 23, no. 5, pp. 3950– 3963, 2024.
- [3] R. Zhao, Y. Ran, J. Luo, and S. Chen, "Towards coverage-aware cooperative video caching in leo satellite networks," in *Pro. IEEE Glob. Commun. Conf. (GLOBECOM)*, pp. 1893–1898, 2022.
- [4] Y. Liu, T. Ma, Z. Tang, X. Qin, H. Zhou, and X. Shen, "Ultra-dense leo satellite access network slicing: A deep reinforcement learning approach," in *Pro. IEEE Glob. Commun. Conf. (GLOBECOM)*, pp. 5043– 5048, 2023.
- [5] G. Xu, F. Tan, Y. Ran, Y. Zhao, and J. Luo, "Joint beam-hopping scheduling and coverage control in multibeam satellite systems," *IEEE Wireless Commun. Lett.*, vol. 12, no. 2, pp. 267–271, 2023.
- [6] J. Choi and V. Chan, "Optimum power and beam allocation based on traffic demands and channel conditions over satellite downlinks," *IEEE Trans. Wireless Commun.*, vol. 4, no. 6, pp. 2983–2993, 2005.
- [7] J. Du, C. Jiang, H. Zhang, X. Wang, Y. Ren, and M. Debbah, "Secure satellite-terrestrial transmission over incumbent terrestrial networks via cooperative beamforming," *IEEE J. Sel. Areas Commun.*, vol. 36, no. 7, pp. 1367–1382, 2018.
- [8] J. e. a. Lei, "Multibeam satellite frequency/time duality study and capacity optimization," *J. Commun. Netw.*, vol. 13, no. 5, pp. 472–480, 2011.
- [9] Y. Su, Y. Liu, Y. Zhou, J. Yuan, H. Cao, and J. Shi, "Broadband leo satellite communications: Architectures and key technologies," *IEEE Wireless Commun.*, vol. 26, no. 2, pp. 55–61, 2019.
- [10] J. Zhu, Y. Sun, and M. Peng, "Timing advance estimation in low earth orbit satellite networks," *IEEE Trans. Veh. Technol.*, vol. 73, no. 3, pp. 4366–4382, 2024.
- [11] S. Yuan, Y. Sun, and M. Peng, "Joint network function placement and routing optimization in dynamic software-defined satellite-terrestrial integrated networks," *IEEE Trans. Wireless Commun.*, vol. 23, no. 5, pp. 5172–5186, 2024.
- [12] V. D. Tuong, N.-N. Dao, W. Noh, and S. Cho, "Deep reinforcement learning-based hierarchical time division duplexing control for dense wireless and mobile networks," *IEEE Trans. Wireless Commun.*, vol. 20, no. 11, pp. 7135–7150, 2021.
- [13] F. Li, K.-Y. Lam, M. Jia, K. Zhao, X. Li, and L. Wang, "Spectrum optimization for satellite communication systems with heterogeneous user preferences," *IEEE Syst. J.*, vol. 14, no. 2, pp. 2187–2191, 2020.
- [14] Y. Kawamoto, T. Kamei, M. Takahashi, N. Kato, A. Miura, and M. Toyoshima, "Flexible resource allocation with inter-beam interference in satellite communication systems with a digital channelizer," *IEEE Trans. Wireless Commun.*, vol. 19, no. 5, pp. 2934–2945, 2020.
- [15] Z. Lin, M. Lin, J.-B. Wang, T. de Cola, and J. Wang, "Joint beamforming and power allocation for satellite-terrestrial integrated networks with non-orthogonal multiple access," *IEEE J. Sel. Topics Signal Process.*, vol. 13, no. 3, pp. 657–670, 2019.
- [16] P. Gu, R. Li, C. Hua, and R. Tafazolli, "Dynamic cooperative spectrum sharing in a multi-beam leo-geo co-existing satellite system," *IEEE Trans. Wireless Commun.*, vol. 21, no. 2, pp. 1170–1182, 2022.
- [17] T. S. Abdu, S. Kisseleff, E. Lagunas, and S. Chatzinotas, "Flexible resource optimization for geo multibeam satellite communication system," *IEEE Trans. Wireless Commun.*, vol. 20, no. 12, pp. 7888–7902, 2021.

- [18] D. Zhou, M. Sheng, Y. Wang, J. Li, and Z. Han, "Machine learningbased resource allocation in satellite networks supporting internet of remote things," *IEEE Trans. Wireless Commun.*, vol. 20, no. 10, pp. 6606– 6621, 2021.
- [19] Y. Yuan, L. Lei, T. X. Vu, Z. Chang, S. Chatzinotas, and S. Sun, "Adapting to dynamic leo-b5g systems: Meta-critic learning based efficient resource scheduling," *IEEE Trans. Wireless Commun.*, vol. 21, no. 11, pp. 9582–9595, 2022.
- [20] L. Yin, R. Yang, Y. Yang, L. Deng, and S. Li, "Beam pointing optimization based downlink interference mitigation technique between ngso satellite systems," *IEEE Wireless Commun. Lett.*, vol. 10, no. 11, pp. 2388–2392, 2021.
- [21] V.-P. Bui, T. Van Chien, E. Lagunas, J. Grotz, S. Chatzinotas, and B. Ottersten, "Joint beam placement and load balancing optimization for non-geostationary satellite systems," in *IEEE Int. Mediterranean Conf. Commun. Netw.*, pp. 316–321, 2022.
- [22] M. Takahashi, Y. Kawamoto, N. Kato, A. Miura, and M. Toyoshima, "Dbf-based fusion control of transmit power and beam directivity for flexible resource allocation in hts communication system toward b5g," *IEEE Trans. Wireless Commun.*, vol. 21, no. 1, pp. 95–105, 2022.
- [23] A. Wang, L. Lei, E. Lagunas, A. I. Pérez-Neira, S. Chatzinotas, and B. Ottersten, "Joint optimization of beam-hopping design and nomaassisted transmission for flexible satellite systems," *IEEE Trans. Wireless Commun.*, vol. 21, no. 10, pp. 8846–8858, 2022.
- [24] X. Hu, Y. Zhang, X. Liao, Z. Liu, W. Wang, and F. M. Ghannouchi, "Dynamic beam hopping method based on multi-objective deep reinforcement learning for next generation satellite broadband systems," *IEEE Trans. Broadcast.*, vol. 66, no. 3, pp. 630–646, 2020.
- [25] J. Zhao, X. Xiong, and Y. Chen, "Design and application of a network planning system based on digital twin network," *IEEE J. Radio Freq. Identif.*, vol. 6, pp. 900–904, 2022.
- [26] M. Jafari, A. Kavousi-Fard, T. Chen, and M. Karimi, "A review on digital twin technology in smart grid, transportation system and smart city: Challenges and future," *IEEE Access*, vol. 11, pp. 17471–17484, 2023.
- [27] Z. Ji, S. Wu, and C. Jiang, "Cooperative multi-agent deep reinforcement learning for computation offloading in digital twin satellite edge networks," *IEEE J. Sel. Areas Commun.*, vol. 41, no. 11, pp. 3414–3429, 2023.
- [28] L. Wang, X. Hu, S. Ma, S. Xu, and W. Wang, "Dynamic beam hopping of multi-beam satellite based on genetic algorithm," in *Proc. IEEE Wireless Commun. Networking Conf. (WCNC)*, pp. 1364–1370, 2020.
- [29] M. Takahashi, Y. Kawamoto, N. Kato, A. Miura, and M. Toyoshima, "Adaptive power resource allocation with multi-beam directivity control in high-throughput satellite communication systems," *IEEE Wireless Commun. Lett.*, vol. 8, no. 4, pp. 1248–1251, 2019.
- [30] A. Wang, L. Lei, E. Lagunas, S. Chatzinotas, A. I. P. Neira, and B. Ottersten, "Joint beam-hopping scheduling and power allocation in nomaassisted satellite systems," in 2021 IEEE Wireless Communications and Networking Conference (WCNC), pp. 1–6, 2021.
- [31] X. Du, X. Hu, Y. Wang, and W. Wang, "Dynamic resource allocation for beam hopping satellites communication system: An exploration," in *Proc. IEEE International Conference on Trust, Security and Privacy in Computing and Communications (TrustCom)*, pp. 1296–1301, 2022.
- [32] L. Wang, C. Zhang, D. Qu, and G. Zhang, "Resource allocation for beam-hopping user downlinks in multi-beam satellite system," in *Proc.* 15th Int. Wireless Commun. Mobile Comput. Conf. (IWCMC), pp. 925– 929, 2019.
- [33] Z. Lin, Z. Ni, L. Kuang, C. Jiang, and Z. Huang, "Dynamic beam pattern and bandwidth allocation based on multi-agent deep reinforcement learning for beam hopping satellite systems," *IEEE Trans. Veh. Technol.*, vol. 71, no. 4, pp. 3917–3930, 2022.
- [34] S. Shi, G. Li, Z. Li, H. Zhu, and B. Gao, "Joint power and bandwidth allocation for beam-hopping user downlinks in smart gateway multibeam satellite systems," *International Journal of Distributed Sensor Networks*, vol. 13, no. 5, p. 1550147717709461, 2017.
- [35] M. G. Kibria, E. Lagunas, N. Maturo, D. Spano, and S. Chatzinotas, "Precoded cluster hopping in multi-beam high throughput satellite systems," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, pp. 1– 6, 2019.
- [36] C. Zhang, X. Zhao, and G. Zhang, "Joint precoding schemes for flexible resource allocation in high throughput satellite systems based on beam hopping," *China Commun.*, vol. 18, no. 9, pp. 48–61, 2021.
- [37] L. Chen, E. Lagunas, S. Chatzinotas, and B. Ottersten, "Satellite broadband capacity-on-demand: Dynamic beam illumination with selective precoding," in *Proc. 29th Eur. Signal Process. Conf. (EUSIPCO)*, pp. 900–904, 2021.

- [38] A. I. Aravanis, B. Shankar M. R., P.-D. Arapoglou, G. Danoy, P. G. Cottis, and B. Ottersten, "Power allocation in multibeam satellite systems: A two-stage multi-objective optimization," *IEEE Trans. Wireless Commun.*, vol. 14, no. 6, pp. 3171–3182, 2015.
- [39] L. Chen, L. Wu, E. Lagunas, A. Wang, L. Lei, S. Chatzinotas, and B. Ottersten, "Joint power allocation and beam scheduling in beam-hopping satellites: A two-stage framework with a probabilistic perspective," *IEEE Trans. Wireless Commun.*, pp. 1–1, 2024.
- [40] W. Liu, F. Tian, Z. Jiang, G. Li, and Q. Jiang, "Beam-hopping based resource allocation algorithm in leo satellite network," in *Space Information Networks*, pp. 113–123, 2019.
- [41] F. Tian, L. Huang, G. Liang, X. Jiang, S. Sun, and J. Ma, "An efficient resource allocation mechanism for beam-hopping based leo satellite communication system," in *Proc. IEEE Int. Symp. Broadband Multimedia Syst. Broadcast.*, pp. 1–5, 2019.
- [42] Z. Shuang, Z. Xing, W. Peng, and W. Wenbo, "Joint beam scheduling and power optimization for beam hopping leo satellite systems," *China Commun.*, pp. 1–14, 2024.
- [43] T. Li, R. Yao, Y. Fan, X. Zuo, and L. Jiang, "Multiobjective optimization for beam hopping and power allocation in dual satellite cooperative transmission networks," *IEEE Syst. J.*, vol. 17, no. 3, pp. 3870–3881, 2023.
- [44] X. Deng, L. Zhu, Q. Tan, Y. Yang, and Y. Zhang, "Multi-layer satellite network resource management based on genetic algorithm," in 2021 Int. Symp. Networks Comput. Commun, pp. 1–6, 2021.
- [45] Z. Lin, Z. Ni, L. Kuang, C. Jiang, and Z. Huang, "Multi-satellite beam hopping based on load balancing and interference avoidance for ngso satellite communication systems," *IEEE Trans. Commun.*, vol. 71, no. 1, pp. 282–295, 2023.
- [46] Z. Lin, Z. Ni, L. Kuang, C. Jiang, and Z. Huang, "Ngso satellites beam hopping strategy based on load balancing and interference avoidance for coexistence with gso systems," *IEEE Commun. Lett.*, vol. 27, no. 1, pp. 278–282, 2023.
- [47] S. Yuan, Y. Sun, M. Peng, and R. Yuan, "Joint beam direction control and radio resource allocation in dynamic multi-beam leo satellite networks," *IEEE Trans. Veh. Technol.*, vol. 73, no. 6, pp. 8222–8237, 2024.